Author's response to reviews

Title: Signs and symptoms of temporomandibular joint disorders related to the degree of mouth opening and hearing loss.

Authors:

Panagiotis Kitsoulis (pkitsoulis@hotmail.com)
Aikaterini Marini (katerinaki88@yahoo.gr)
Kalliopi Iliou (kallina2@hotmail.com)
Vasiliki Galani (vgalani@cc.uoi.gr)
Aristides Zimpis (info@ahzimbis.gr)
Panagiotis Kanavaros (pkanavar@cc.uoi.gr)
Georgios Paraskevas (g_paraskevas@yahoo.gr)

Version: 6 Date: 21 May 2011

Author's response to reviews: see over
Dear Sirs,

Here we submit the revised final form of our manuscript. We have done all the changes that you suggested. More precisely:

1. We corrected all the typing errors throughout the text.

2. Reference 63 was by mistake on page 8 as this reference has been erased in previous versions of the manuscript. We deleted it.

3. All the tables are now numbered as 1-10 without intra-divisions. As 1-10 are now presented in the text.

4. All the coloured parts of the text has been removed and turned into black.

5. All the vertical lines of the tables have been removed.

Thank you very much for the co-operation and we remain at your disposal for any further corrections.

Yours Sincerely,

Prof. P. Kitsoulis